Nodule Worm Infection in Humans and Wild Primates in Uganda: Cryptic Species in a Newly Identified Region of Human Transmission

Ria R. Ghai¹, Colin A. Chapman^{2,3}, Patrick A. Omeja³, T. Jonathan Davies¹, Tony L. Goldberg^{3,4}*

1 Department of Biology, McGill University, Montreal, Quebec, Canada, 2 Department of Anthropology and McGill School of Environment, Montreal, Quebec, Canada, and Wildlife Conservation Society, Bronx, New York, New York, United States of America, 3 Makerere University Biological Field Station, Fort Portal, Uganda, 4 Department of Pathobiological Sciences, School of Veterinary Medicine, University of Wisconsin-Madison, Madison, Wisconsin, United States of America

Abstract

Introduction: Soil-transmitted helminths (STHs) are a major health concern in tropical and sub-tropical countries. *Oesophagostomum* infection is considered endemic to West Africa but has also been identified in Uganda, East Africa, among primates (including humans). However, the taxonomy and ecology of *Oesophagostomum* in Uganda have not been studied, except for in chimpanzees (*Pan troglodytes*), which are infected by both *O. bifurcum* and *O. stephanostomum*.

Methods and Findings: We studied *Oesophagostomum* in Uganda in a community of non-human primates that live in close proximity to humans. Prevalence estimates based on microscopy were lower than those based on polymerase chain reaction (PCR), indicating greater sensitivity of PCR. Prevalence varied among host species, with humans and red colobus (*Procolobus rufomitratus*) infected at lowest prevalence (25% and 41% by PCR, respectively), and chimpanzees, olive baboons (*Papio anubis*), and l'hoest monkeys (*Cercopithecus lhoesti*) infected at highest prevalence (100% by PCR in all three species). Phylogenetic regression showed that primates travelling further and in smaller groups are at greatest risk of infection. Molecular phylogenetic analyses revealed three cryptic clades of *Oesophagostomum* that were not distinguishable based on morphological characteristics of their eggs. Of these, the clade with the greatest host range had not previously been described genetically. This novel clade infects humans, as well as five other species of primates.

Conclusions: Multiple cryptic forms of *Oesophagostomum* circulate in the people and primates of western Uganda, and parasite clades differ in host range and cross-species transmission potential. Our results expand knowledge about human *Oesophagostomum* infection beyond the West African countries of Togo and Ghana, where the parasite is a known public health concern. *Oesophagostomum* infection in humans may be common throughout Sub-Saharan Africa, and the transmission of this neglected STH among primates, including zoonotic transmission, may vary among host communities depending on their location and ecology.

Citation: Ghai RR, Chapman CA, Omeja PA, Davies TJ, Goldberg TL (2014) Nodule Worm Infection in Humans and Wild Primates in Uganda: Cryptic Species in a Newly Identified Region of Human Transmission. PLoS Negl Trop Dis 8(1): e2641. doi:10.1371/journal.pntd.0002641

Editor: Aaron R. Jex, University of Melbourne, Australia

Received July 29, 2013; Accepted November 28, 2013; Published January 9, 2014

Copyright: © 2014 Ghai et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was funded by the Natural Sciences and Engineering Research Council of Canada (Discovery Grant: CAC, and Alexander Graham Bell Award: RRG) and the Fonds de Recherche du Quebec Nature et Technologies International Internship award, as well as by NIH grant TW009237 as part of the joint NIH-NSF Ecology of Infectious Disease program and the UK Economic and Social Research Council. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: tgoldberg@vetmed.wisc.edu

Introduction

Soil-transmitted helminths (STHs) are parasitic nematodes that cause infection via eggs and larvae, which are shed in feces and persist in the soils of tropical and sub-tropical countries [1]. STHs infect over one billion people worldwide [2] and may cause a combined disease burden as substantial as that caused by malaria or tuberculosis [3]. Nevertheless, these parasites are largely neglected in research, perhaps in part because the diseases they cause are suffered by the world's most impoverished populations [1]. Although roundworms (Ascaris lumbricoides), hookworms (Necator americanus and Ancylostoma duodenale) and whipworms (Trichuris trichiura) are of global importance, other "lesser" parasites are localized to specific regions [1]. This includes Oesophagostomum spp., a genus of nodule-causing worms with L3 larvae that are infective *via* ingestion after 4–7 days [4–7]. The human burden of *Oeosphagostostomum* infection is considered localized to West Africa, specifically the countries of Togo and Ghana [5,8,9].

A variety of mammals, including pigs, ruminants [10,11], and non-human primates are frequently parasitized by *Oesophagostomum*. Infections in wild primates appear to be asymptomatic; clinical signs and mortality due to *Oesophagostomum* have only been recorded in captive settings [10,12]. Eight species of *Oesophagostomum* have been recorded in wild primates, of which the three most common, *O. bifurcum*, *O. stephanostomum*, and *O. aculeatum*, are able to infect humans [5,7,11,13]. Of these, *O. bifurcum* appears to be the only species to regularly parasitize humans, with human infections by other species considered incidental [5,9]. In Togo

Author Summary

Nodule worms infect the gastrointestinal tracts of a number of mammalian species, including humans and other primates. This study sought to identify the species of nodule worms causing infections within and around an East African national park in Uganda where monkeys and apes co-occur and overlap with people. Some primates, particularly those traversing large distances in small groups, were most susceptible to nodule worm infection. Additionally, molecular analyses identified three separate groups of nodule worm that could not be distinguished based on microscopic examination of their eggs. One of these groups was found in humans as well as other primates and had not previously been genetically characterized. These results suggest that certain types of nodule worm may be restricted to particular hosts, while others may be transmitted among primates, including humans. Nodule worms are currently thought to be a human health concern only in some West African countries. This research suggests that nodule worms have a broader geographic impact in humans than previously appreciated.

and Ghana, the majority of human *Oesophagostomum* cases occur within endemic foci [5,8] and affect 20% and 90% of the population, respectively, with prevalence highest in rural areas [12,14,15]. The only known species to cause infection within these countries is *O. bifurcum*, which also infects the region's non-human primates, including patas monkeys (*Erythrocebus patas*), mona monkeys (*Cercopithecus mona*), and olive baboons (*Papio anubis*) [4,10,16]. However, previous research has indicated that *O. bifurcum* is not commonly transmitted among primate species (including humans) because different parasite variants within the species are adapted to specific hosts [4,16,17].

In Uganda, a number of primate species harbor *Oesophagosto*mum, as evidenced by microscopic detection of eggs in feces. These include members of the primate subfamilies Cercopithecinae and Colobinae, as well as chimpanzees (*Pan trogolodytes*) [18–20]. There have also been reports of oesophagostomiasis in human patients in Uganda, although no such reports, to our knowledge, have been published since the 1980s [5,21], perhaps due to under-reporting or improvements in treatment. With the exception of chimpanzees, which are infected with both *O. bifurcum* and *O. stephanostomum* [22], the species of *Oesophagostomum* infecting Ugandan primates and humans remains unknown.

In this study, we examined Oesophagostomum within the primate community of Kibale National Park, Uganda, using a combination of microscopic and molecular methods. Species-specific identification of eggs by microscopy alone is difficult, because eggs are similar morphologically to other STHs, including hookworms, Trichostrongylus spp., and the "false hookworm" Ternidens deminutus [5,9,23-25]. In other studies, coproculture of L3 larvae or necropsy to isolate adult worms have been used to identify these parasites to species [25,26]. Here, we used molecular methods to detect and sequence Oesophagostomum DNA directly from feces; such methods have proven informative for other similar studies [27,28]. In addition, we used phylogenetic comparative methods to ascertain whether primate host traits explain variation in prevalence of Oesophagostomum infection among host species. Our sampling and analyses included nearby human populations to assess whether Oesophagostomum is a public health concern in the region, as well the parasite's local propensity for zoonotic transmission.

Methods

Ethics Statement

Prior to data collection, all protocols were reviewed and approved by the Uganda National Council for Science and Technology and the Uganda Wildlife Authority, as well as by the Institutional Review Board and the Animal Care and Use Committees of McGill University and the University of Wisconsin-Madison. Due to low literacy rates, oral informed consent was obtained from all adult subjects and a parent or guardian of all minor participants by trained local field assistants and documented by witnessed notation on IRB-approved enrollment forms.

Study Site and Sample Collection

Kibale National Park $(0^{\circ}13'-0^{\circ}41'N, 30^{\circ}19'-30^{\circ}32'E)$ is a 795 km² semi-deciduous protected area in Western Uganda. Primate research has occurred in Kibale for over four decades, focusing on chimpanzees and red colobus monkeys (*Procolobus rufomitratus*) [29,30]. As a result, a number of primate groups are habituated to human presence, and many individuals are recognizable based on a combination of physical attributes and collars affixed as part of a larger project on primate health and conservation [31].

Samples from monkeys in the Kanyawara area of Kibale National Park were collected from red-tailed guenons (Cercopithecus ascanius), blue monkeys (Cercopithecus mitis), l'hoest monkeys (Cercopithecus lhoesti), grey-cheeked mangabeys (Lophocebus albigena), olive baboons (Papio anubis), red colobus, and black and white colobus (Colobus guereza) (Figure 1). Chimpanzee samples were collected from Kanyanchu, an area that has a habituated chimpanzee community as a result of tourism (Figure 1). All samples were collected non-invasively immediately after defecation and placed into sterile tubes. Date, location, species, age and sex category, and social group membership were recorded. Human samples were collected after the receipt of Institutional Review Board-approved informed consent following World Health Organisation protocols. Samples collection occurred in three villages: Ibura, Kanyansohera, and Kasojo, which are less than 5 km from the border of the park (Figure 1). Individuals between the ages of 2 and 70 were suitable participants of this study. Consenting participants were given instructions on how to collect the sample, which was then retrieved for processing within one day.

Samples were subjected to a modified ethyl acetate concentration method, recommended in the approved guidelines of the Clinical and Laboratory Standards Institute for the identification of intestinal-tract parasites [32,33]. Concentration by sedimentation was performed in the field using one gram of undiluted feces without fixture in formalin, as formalin is a known inhibitor of the polymerase chain reaction [34]. All materials were sterilized prior to use, and care was exercised throughout the procedure to prevent contamination. Sediments were left uncapped for two hours after completion of the procedure to allow ethyl acetate that may inhibit polymerase chain reaction (PCR) to volatilize. Sediments were then suspended in 2 mL RNALater nucleotide stabilization solution (Sigma-Aldrich, St. Louis, MO, USA) and frozen at -20° C until shipment to North America.

Microscopy

Thin smears from sedimented feces were used for microscopy [35]. All eggs of the genus *Oesophagostomum* were identified at $10 \times$ objective magnification on a Leica DM2500 light microscope. Data were recorded on size, shape, color and internal contents of eggs. Images were captured at $40 \times$ objective magnification of all specimens using an Infinity1 CMOS digital microscope camera



Figure 1. Map of sample collection sites in and around Kibale National Park, Uganda. doi:10.1371/journal.pntd.0002641.g001

and Infinity Camera v.6.2.0 software (Lumenera Corporation, Ottawa, ON, Canada). Samples were considered negative after the entire sediment sample was scanned and no eggs were found. We note that while identification of *Oesophagostomum* eggs was based on a rigorous set of characteristics, this genus cannot easily be distinguished from hookworm infection by eggs alone. However, hookworms have not been found in previous surveys of the gastrointestinal parasites of this primate community [19,20], suggesting that eggs identified with morphological characteristics of both *Oesophagostomum* and hookworm were almost certainly *Oesophagostomum*.

Molecular Methods

DNA was extracted from 200 µL of sedimented feces using a ZR Fecal DNA MiniPrep Kit (Zymo Research Corporation, Irvine, CA, USA), following manufacturer protocols. External PCR was performed targeting the ribosomal internal transcribed spacer 2 gene using primers NC1 (5'-ACGTCTGGTT-CAGGGTTGTT-3') and NC2 (5'-TTAGTTTCTTTTCCTCC-GCT-3'), which generated products that ranged in size from 280 to 400 bp, suggesting that, as expected, the primer set detected a number of parasitic helminths present in the samples [27,36]. Subsequently, an internal, semi-nested PCR generating amplicons of predicted size 260 bp was performed using primer NC2 and newly designed Oesophagostomum-specific primer, OesophITS2-21 (5'-TGTRACACTGTTTGTCGAAC-3'). Primer OesophITS2-21 was generated by aligning publicly available sequences of the Oesophagostomum internal transcribed spacer 2 gene [26,36-39], and GenBank accession numbers HQ283349, HQ844232]. In total, eight species of *Oesophagostomum* were represented in the alignment. Other species of varying relatedness, including other members of the taxa Chabertiidae (*Chabertia ovina*, Accession No. JF680981; *Ternidens deminutus*, Accession No. HM067975), Strongylidae (*Strongylus vulgaris* [40]), and Strongylida (*Necator americanus* [36], and *Ancylostoma duodenale* [41]) were also included. Priming regions were selected to be identical among all species of *Oesophagostomum* but divergent from the other genera. Primer ITS2-21 was highly specific as confirmed by sequencing, since all PCR products matched *Oesophagostomum* despite the fact that a number of other parasites (including *Strongyloides, Necator* and *Trichuris*), were identified in the same samples during microscopic examination.

External PCR was performed in 25 µL volumes using the FailSafe System (Epicentre Biotenchnologies, Madison, WI, USA) with reactions containing 1× FailSafe PCR PreMix with Buffer C, 1 Unit of FailSafe Enyme Mix, 2.5 picomoles of each primer (NC1 and NC2), and 1 µL of template. Reactions were cycled in a Bio-Rad CFX96 platform (Bio-Rad Laboratories, Hercules, CA, USA) with the following temperature profile: 94°C for 1 min; 45 cycles of 94°C for 15 sec, 50°C for 30 sec, 72°C for 90 sec; and a final extension at 72°C for 10 min. Internal PCR was performed in 25 µL volumes using the DyNAzyme DNA Polymerase Kit (Thermo Scientific, Asheville, NC, USA) with reactions containing 0.5 Units of DyNAzyme I DNA Polymerase, 1× Buffer containing 1.5 mM MgCl₂, 2.5 picomoles of each primer (OesophITS2-21 and NC2), and 1 µL of template. Reactions were cycled with the following temperature profile: 95°C for 1 min; 45 cycles of 95°C for 15 sec, 55°C for 30 sec, 70°C for 90 sec; and a final extension at 70°C for 5 min. Amplicons were electrophoresed on 1% agarose gels stained with ethidium bromide, and purified from gels using the Zymoclean Gel DNA Recovery Kit (Zymo Research Corporation, Irvine, CA, USA) according to the manufacturer's instructions.

Products were Sanger sequenced in both directions using primers OesophITS2-21 and NC2 on ABI 3730xl DNA Analyzers (Applied Biosystems, Grand Island, NY, USA) at the University of Wisconsin-Madison Biotechnology Center DNA Sequencing Facility. Sequences were hand-edited and assembled using Sequencher v4.9 (Gene Codes Corporation, Ann Arbor, MI, USA) and all ambiguous bases were resolved by repeat PCR and re-sequencing, as described above. All new sequences were deposited in GenBank, under Accession Numbers KF250585 - KF250660.

Phylogenetic Analyses

Sequences were aligned using the computer program ClustalX [42] with minor manual adjustment. Published reference sequences were included to identify putative species (AF136575, Y11733, AF136576) and as outgroups (HQ844232, Y11738, Y11735, Y10790, AJ006149), and were trimmed to the length of the newly generated sequences using Mesquite v.2.75 [43]. Trimmed sequences vielded the same tree topology as did untrimmed sequences (by neighbor-joining method; results not shown), suggesting that the amplified region was sufficient for taxonomic discrimination. Phylogenetic trees were reconstructed using maximum likelihood in MEGA v.5.05 [44] and the Hasegawa-Kishino-Yano substitution model [45]. Phylogenetic support was assessed using 1,000 bootstrap replicates. To estimate Oesophagostomum genetic diversity, percent nucleotide-level sequence identity among sequences was calculated as the uncorrected pairwise proportion of nucleotide differences (p-distance) in MEGA v5.05 [44].

Statistical Analysis

Diagnostic performance of microscopy versus PCR was estimated by calculating sensitivity (*i.e.*, true positive rate) and specificity (i.e., true negative rate) using MedCalc v.12.5.0 (MedCalc Software, Ostend, Belgium). Prevalence of infection was calculated as the number of samples found to be positive for Oesophagostomum divided by the total number of samples collected, with 95% confidence intervals calculated using the modified Wald method [46]. To determine whether prevalence differed among primate host species, a chi-square test was conducted in Quantitative Parasitology v3.0 [47]. To explore variation in prevalence among hosts while controlling for their phylogenetic non-independence, a phylogenetic least squared regression (PGLS) was conducted in R [48] using the ape [49] and caper [50] libraries. Prevalence of Oesophagostomum was included as the dependent variable, and various primate life history traits were independent variables: terrestriality (predominantly terrestrial versus predominantly arboreal), maximum home range [51–56], maximum group size [51,53,55,57-60], percentage time spent in polyspecific associations [61,62], average female body mass, and average daily travel distance (the latter was log transformed since the relationship was close to exponential) [55,56,62-64]. Humans were omitted from the PGLS analysis because many of these traits vary widely among human populations, making accurate estimations problematic.

To determine the degree to which each *Oesophagostomum* lineage (*i.e.*, taxonomic unit) identified by DNA sequencing was host restricted, we calculated the phylogenetic dispersion of infected hosts using the net relatedness index (NRI) in R [48] using the ape [49] and picante libraries. Mean pairwise distance (MPD) was weighted by the ratio of occurrence of each *Oesophagostomum* within each lineage, and compared to null expectation in 1000 randomly assembled communities. Results are reported as standard effects sizes, with values close to 1 indicating phylogenetic evenness (*i.e.*, *Oesophagostomum* lineages infect a greater diversity of hosts than would be expected by chance), while values <0.05 indicate phylogenetic clustering (*i.e.*, *Oesophagostomum* lineages are host-specific).

Results

A total of 318 fecal samples from primates, including humans, were collected (Table 1). Of these, 112 were positive for *Oesophagostomum* by microscopy, for a community-wide prevalence of infection of 35.2% (Table 1). All eggs identified by microscopy were similar in internal and external morphology in samples from all primate species (Figure 2). Eggs were 65-80 by $35-50 \mu m$ in size, which is consistent with previous results from this community [19,20] (Figure 2).

PCR generated single, clear amplicons of expected size (260 bp) in 222 samples, indicating positive detection of *Oesophagostomum* DNA, for an overall prevalence of 69.8%. No amplicons were present in remaining samples. Resulting DNA sequences overlapped 100% with published sequences and contained no insertions or deletions, making alignment unequivocal.

When PCR results were compared to microscopy, the overall sensitivity of PCR was 100% (95% CI 96.8%–100.0%), but specificity was only 47.5% (95% CI 40.5%–54.7%). Thus, PCR did not classify any microscopy-positive samples as negative but identified 109 microscopy-negative samples as positive.

Prevalence of *Oesophagostomum* infection (as determined by both microscopy and PCR) varied significantly among host species (microscopy: chi-square = 54.31, df = 8, P<0.0001; PCR: chi-square = 112.2, df = 8, P<0.0001). Both microscopy and PCR identified humans as having the lowest prevalence of infection (8.3% and 25.0%, respectively), followed by red colobus (17.2% and 40.6%, respectively). Chimpanzees, l'hoest monkeys, and olive baboons had the highest prevalence by both methods, with 100%

prevalence by PCR in all three species (although sample sizes were low in some cases; Table 1).

PGLS indicated that terrestriality, maximum home range, maximum group size, percent time spent in polyspecific associations, and average female body mass were not significant univariate predictors of *Oesophagostomum* prevalence (all P>0.05 from PGLS with lambda = ML; Table 2). However, log daily travel explained nearly 55% of the variation in prevalence among host species (P<0.05, $R^2=0.546$, from PGLS with the ML estimate of lambda = 0). In a multivariate model, both group size and log daily travel were significant predictors of prevalence, with group size showing a negative relationship and log daily travel a positive relationship (Table 2). This two-predictor model including group size and daily travel explained over 75% of the variation in *Oesophagostomum* prevalence among species (model P<0.01, $R^2=0.7701$; Table 3).

From 222 positive samples, 76 were randomly selected for sequencing to represent as even a number of positive samples per host species as possible. All 76 sequences most closely matched published Oesophagostomum ITS-2 DNA sequences using the BLASTn tool on the National Centre for Biotechnology Information website. Phylogenetic analysis resolved these sequences into three clades (Figure 3). Clade 1 contained all 12 sequences from olive baboons, one sequence from l'hoest monkeys, one sequence from grev-cheeked mangabeys, three sequences from red colobus and one sequence from red-tailed guenons. These sequences were identical to published reference sequences for O. bifurcum [26,36]. Five additional sequences from l'hoest monkeys sorted into clade 1 and were 97.1% similar to this same O. bifurcum reference sequence. Clade 2 contained all eight sequences from chimpanzees, five sequences from blue monkeys, two sequences from black and white colobus, two sequences from grey-cheeked mangabeys, three sequences from red colobus, and twelve sequences from red-tailed guenons. All sequences in clade 2 were identical to an O. stephanostomum reference sequence [26]. Clade 3 was composed of two nearly identical branches (99.4% identity) that contained all six sequences from humans, as well as sequences from three blue monkeys, three black and white colobus, five grevcheeked mangabeys, two red colobus, and two red-tailed guenons. These sequences were 92.4-93.0% and 93.0-93.6% similar to O. *bifurcum*, and *O. stephanostomum*, respectively, but were not identical to any published reference sequence.

Host species were phylogenetically clustered within *O. bifurcum* clade 1 (NRI = -1.76, P < 0.05). Clade 2 (*O. stephanostomum*) did not vary significantly from the null expectation of no host clustering, NRI = 0.86, P = 0.75). Clade 3 was marginally phylogenetically over-dispersed with respect to distribution of host species (NRI = 1.24, P = 0.04).

Discussion

Here we evaluate the prevalence of *Oesophagostomum* infection in wild primates and humans in Western Uganda using both microscopy and PCR. Our results clearly show that prevalence varied significantly among host species. Humans had the lowest prevalence of infection likely because of avoidance behaviors such as sanitation practices [65,66] and because of the common use of antihelminthics in the region. Red colobus and black and white colobus also had comparatively low prevalence of infection, as found in previous studies [16,20,67]. This observation may reflect colobine gastrointestinal physiology, which is characterized by folivory and foregut fermentation [68], and the associated regular ingestion of plant secondary compounds that may suppress infection by pathogenic organisms [69]. Conversely, the high



Figure 2. Microscopic images of representative *Oesophagostomum* sp. eggs found in the feces of infected primate hosts. Images were captured at $40 \times$ objective magnification from thin smears of sedimented feces. A = blue monkey, B = black and white colobus, C = chimpanzee, D = l'hoest monkey, E = grey-cheeked mangabey, F = olive baboon, G = red-tailed guenon, H = human. All eggs were between 65 and 84 µm long and between 35 and 55 µm wide. doi:10.1371/journal.pntd.0002641.g002

prevalence of infection in chimpanzees, olive baboons, and l'hoest monkeys may reflect reduced physiological barriers to infection or increased susceptibility. To explain this interspecific variation in prevalence, we examined correlations between life history variables and prevalence among host species. We found that two variables, daily travel distance and group size, explain over 75% of the variance in *Oesophagostomum* prevalence among host species. Surprisingly, body mass, the strongest predictor of helminth species richness elsewhere, was not significant here [67].

Previous studies have concluded that group-living animals with small home ranges are likely to suffer high intensities of infection due to frequent environmental re-exposure [70–72]. Our results indicate

the opposite in the case of *Oesophagostomum*: smaller primate groups with large daily travel distances had higher prevalence. Animals with larger day ranges may encounter greater habitat variation [73], which may increase exposure to *Oesophagostomum* from environmental sources. In addition, previous research has implicated terrestriality as an important factor affecting the prevalence of trematode parasites in primates [74]. In our study, the three host species with highest *Oesophagostomum* prevalence (chimpanzees, olive baboons and l'hoest monkeys) were also the only three predominantly terrestrial species. Although this trend was not statistically significant, it is possible that terrestrial primates contact soil more frequently, and thus the infective stages of STHs.

Table 1. Prevalence of Oesophagostomum spp.	in nine primate host species	، (including humans) in and n	ear Kibale National Park,
Uganda, based on microscopy and PCR.			

		Number positi	ve	Prevalence (95% CI))
Species	N	Microscopy	PCR	Microscopy	PCR
BM (Blue monkey)	33	10	24	30.3 (17–47)	72.7 (56–85)
BW (Black and white colobus)	37	8	21	21.6 (11–37)	56.8 (41–71)
CH (Chimpanzee)	30	18	30	60.0 (42–75)	100 (86–100)
GM (Gray-cheeked mangabey)	42	17	39	40.5 (27–56)	92.9 (80–98)
HU (Human)	36	3	9	8.3 (2–23)	25.0 (14–41)
LM (L'hoest monkey)	8	6	8	75.0 (40–94)	100 (63–100)
OB (Olive baboon)	27	18	27	66.7 (48–81)	100 (85–100)
RC (Red colobus)	64	11	26	17.2 (10–28)	40.6 (29–53)
RT (Red-tailed guenon)	41	21	38	51.2 (36–66)	92.7 (80–98)
TOTAL	318	112	222	35.2 (30–40)	69.8 (65–75)

doi:10.1371/journal.pntd.0002641.t001

Table 2. Phylogenetic generalized least-squared multiple regression models of the relationship between *Oesophagostomum* prevalence and univariate life-history variables of diurnal primates (excluding humans) within the Kibale community.

Univariate Model	λ	Slope	F	Р	Adjusted r ²
Terrestriality	0.97	0.20495	2.93	0.130	0.2159
Home range	1.00	0.00004	1.07	0.400	0.0101
Group size	1.00	-0.00023	0.05	0.950	-0.1567
Polyspecific association	1.00	-0.19713	1.16	0.375	0.0225
Body mass	1.00	0.00400	0.61	0.578	-0.0590
Log daily travel	0.00	0.18500	9.43	0.014	0.5464

doi:10.1371/journal.pntd.0002641.t002

Although group size was not a significant predictor of prevalence in univariate analyses, our multivariate analysis found smaller groups with large daily travel distances to be at greatest risk of infection. This finding contrasts with previous studies showing that increased intragroup contact increases exposure [71,75]. In Kibale, positive associations between group size and parasite richness have been documented for protozoan parasites in mangabeys [76]. However parasite richness is not necessarily associated with prevalence. Small primate groups might maintain high intra-group infection rates for certain parasites if transmission within the group is frequent, thus maintaining high prevalence (as seen here) without correspondingly high parasite richness.

Our study detected substantial cryptic phylogenetic diversity in *Oesophagostomum* infecting Ugandan primates. Currently, the principal human *Oesophagostomum* species is considered to be *O. bifurcum* [5], while other great apes harbor *O. stephanostomum* [6,12,18,77]. Recently, however, chimpanzees inhabiting a northern sector of Kibale were identified as positive for *O. bifurcum*, making this the first discovery of *O. bifurcum* in nonhuman apes. The same study identified chimpanzees also infected or co-infected with *O. stephanostomum* [22]. In our phylogenetic

Table 3. Phylogenetic generalized least-squared multiple regression models of the relationship between *Oesophagostomum* prevalence and life-history variables of diurnal primates (excluding humans) within the Kibale community.

Multivariate Model	λ	Slope	t	<i>P</i> trait	F	P overall	Adjusted r ²
y-intercept	0.00	-1.93390	-2.65	0.045	9.95	0.015	0.7189
Home range +		0.00012	-2.16	0.083			
Log daily travel		0.00039	3.70	0.014			
y-intercept	0.00	-0.89378	-2.58	0.049	12.73	30.009	0.7701
Group size +		-0.00180	-2.61	0.047			
Log daily travel		0.24894	5.04	0.004			
y-intercept	0.00	-1.48140	-1.97	0.120	8.39	0.032	0.7599
Home range +		-0.00006	-0.88	0.425			
Group size +		-0.00127	-1.36	0.245			
Log daily travel		0.33085	3.14	0.035			

doi:10.1371/journal.pntd.0002641.t003

analysis, we identified both *O. bifurcum* and *O. stephanostomum* in the Kibale primate community. However, we found only *O. stephanostomum* in chimpanzees, although the possibility of undetected *O. bifurcum* infections cannot be ruled out.

In addition, we identified a third Oesophagostomum lineage that did not cluster with any published sequence and thus may represent a previously uncharacterized taxon. It is possible that this new taxon has remained undetected in previous molecular investigations. We examined the OB primer that has been used previously to identify O. bifurcum [36] and conclude that it would probably not amplify our newly identified taxon due to mismatched bases at both the 5' and 3' ends of the primer. It is therefore possible that the new taxon we identified exists elsewhere (e.g. in Togo and Ghana) but has been not been detected or differentiated from other members of the genus. However, we caution that these inferences are based on a short region of a single gene, and that sequencing additional genes as well as morphological characterization of L3 larvae and adults will be necessary to confirm these findings. Nonetheless, our results suggest a heretofore unappreciated degree of hidden genetic diversity within this well-described genus of parasites that are known to infect humans.

Interestingly, all Oesophagostomum sequences recovered from humans clustered with the previously undescribed third taxon, and not with published O. bifurcum sequence from humans elsewhere in Africa (clade 1) [36]. In Ghana, geographic separation between humans and non-human primates infected with Oesophagostomum, despite apparently conducive environments for zoonotic transmission, motivated efforts to determine the host range of the parasite using molecular methods [28]. Genome-wide analyses (amplified fragment length polymorphism, random amplification of polymorphic DNA) suggested that O. bifurcum clusters into distinct groups by host species, thus suggesting that zoonotic transmission is uncommon [4,17]. By contrast, in our study area, no such geographic separation exists between humans and non-human primates. In this setting, we found that both humans and non-human primates were infected with the novel Oesophagostomum clade 3, which is phylogenetically over-dispersed compared to the other Oesophagostomum clades. While our conclusions await verification from more detailed examination of the Oesophagostomum genome, our results nevertheless suggest that this novel clade may be broadly transmissible among species of distantly related primate hosts, including humans. The Kibale ecosystem is known for its high degree of spatio-temporal overlap between humans and non-human primates and its ensuing high rates of transmission of diverse pathogens across primate species [78-81]. Our results provide further evidence for cross-species pathogen transmission between wild primates and humans in this region.

Our paired analyses applying both microscopy and PCR to the same samples indicate that traditional methods based on microscopy may significantly underestimate prevalence. Concentration methods followed by microscopic visualisation of eggs in thin smears are considered definitive diagnostic methods for soiltransmitted helminth (STH) infections [82,83]. Previous studies that have used fecal sedimentation and microscopy have reported *Oesophagostomum* infection prevalence estimates between approximately 3% and 10% in wild Ugandan primates [19,20]. These values are considerably lower than what we report here using molecular methods; however, our results parallel other studies that have estimated prevalence using molecular methods [16,22]. Not surprisingly, we find that PCR is more sensitive than microscopy, perhaps because it can detect *Oesophagostomum* infection even when eggs are not present. For example, tissues or secretions shed by



Figure 3. Phylogenetic analysis of *Oesophagostomum* **based on ITS2 rDNA (260 bp) sequences.** Nucleotide sequences were aligned using Clustal X software [42]. Phylogenetic relationships were inferred in MEGA5 [44], using the maximum likelihood method with a Hasegawa-Kishino-Yano model of nucleotide substitution [45]. The best-scoring maximum-likelihood tree is shown here (-InL = 656.5). Bootstrap values (%) greater than 50% are shown. Taxon names of sequences generated in this study are in bold and correspond to the host species followed by the number of infected individuals in parentheses (BM = blue monkey, BW = black and white colobus, CH = chimpanzee, LM = I'hoest monkey, GM = grey-cheeked mangabey, OB = olive baboon, RC = red colobus, RT = red-tailed guenon, and HU = human). Reference sequences correspond to Genbank accession numbers AF136575 and Y11733 for *O. bifurcum*, AF136576 for *O. stephanostomum*, HQ844232 for *O. sp*, Y11738 for *O. quadrispinulatum*, Y11735 for *O. dentatum*, Y10790 for *O. venulosum*, and AJ006149 for *O. radiatum*. Scale bar indicates nucleotide substitutions per site. doi:10.1371/journal.pntd.0002641.g003

adult worms into the intestinal lumen would be detected by PCR, as would eggs that have hatched into L1 larva prior to fixation during the sedimentation procedure.

To our knowledge, ours is the first study in several decades to report human *Oesophagostomum* infection in Uganda, a country that is over 3,000 km from known foci of infection in West Africa [5]. Given that the prevalence of *Oesophagostomum* was 25% in our sample of people, we suspect that this parasitic infection occurs more commonly across Sub-Saharan Africa than previously thought and may be causing infections that are untreated or misdiagnosed. Our finding of a previously genetically uncharacterized lineage of *Oesophagostomum* that may be transmitted among primate species underscores that the diversity (genetic and otherwise) of this parasite genus may be under-sampled in Africa. Further ecological studies of *Oesophagostomum* in Uganda and elsewhere are needed to quantify the degree of enzootic versus zoonotic transmission. Regardless of the outcome of such research, our results suggest that *Oesophagostomum* should be considered a

References

- Bethony J, Brooker S, Albonico M, Geiger SM, Loukas A, et al. (2006) Soiltransmitted helminth infections: ascariasis, trichuriasis, and hookworm. Lancet 367: 1521–1532.
- W.H.O.. (2004) Report of the third global meeting of the partners for parasite control: deworming for health and development. Geneva, Switzerland: World Health Organization.
- Chan M (1997) The global burden of intestinal nematode infections—fifty years on. Parasitol Today 13: 438–443.
- de Gruijter JM, Ziem J, Verweij JJ, Polderman AM, Gasser RB (2004) Genetic substructuring within *Oesophagostomum bifurcum* (Nematoda) from human and nonhuman primates from Ghana based on random amplified polymorphic DNA analysis. Am J Trop Med Hyg 71: 227–233.

pathogen of concern beyond its accepted foci of infection in Togo and Ghana, and perhaps across all of equatorial Africa.

Acknowledgments

We gratefully acknowledge the Uganda Wildlife Authority and the Uganda National Council for Science and Technology for granting permission to conduct this research. We also thank Dennis Twinomugisha and Lauren Chapman for assistance with logistics, Chesley Walsh, Robert Basaija, Peter Tuhairwe, and Richard Kaseregenyi for assistance in the field, Dwight Bowman for assistance with parasitological identifications, Aleia McCord and Mary Thurber for laboratory training, Johanna Bleecker for map preparation, and Tavis Anderson, Maxwell Farrell, and Chelsey Chisholm for analytical advice.

Author Contributions

Conceived and designed the experiments: RRG CAC TJD TLG. Performed the experiments: RRG CAC PAO TLG. Analyzed the data: RRG TJD TLG. Wrote the paper: RRG TJD TLG.

- Polderman AM, Blotkamp J (1995) Oesophagostomum infections in humans. Parasitol Today 11: 451–456.
- Crestian J, Crespeau F (1975) Oesophagostomum stephanostomum in the chimpanzee. Receuil de medicine Veterinaire de l'Ecole d'Alfort 151: 13–18.
- 7. Chabaud AG, Larivière M (1958) Oesophagostomum parasites in man. Bulletin de la Société de pathologie exotique et de ses filiales 51: 384.
- Polderman AM, Krepel HP, Baeta S, Blotkamp J, Gigase P (1991) Oesophagostomiasis, a common infection of man in Northern Togo and Ghana. Am J Trop Med Hyg 44: 336–344.
- Krepel HP (1994) Oesophagostomum bifurcum infection in man: a study on the taxonomy, diagnosis, epidemiology and drug treatment of Oesophagostomum bifurcum in northern Togo and Ghana [Ph.D.]: Leiden University.

- Skrjabin K, Shikhobalova N, Schulz R, Popova T, Boev S, et al. (1961) Key to parasitic nematodes. Jerusalem: Israel Program for Scientific Translations.
- Krief S, Jamart A, Mahé S, Leendertz FH, Mätz-Rensing K, et al. (2008) Clinical and pathologic manifestation of oesophagostomosis in African great apes: does self-medication in wild apes influence disease progression? J Med Primatol 37: 188–195.
- Blotkamp J, Krepel HP,Baeta V K,VtNJ S, M et al. (1993) Observations on the morphology of adults and larval stages of *Oesophagostomum* sp. isolated from man in northern Togo and Ghana. J Helminthol 67: 49–61.
- Storey PA, Faile G, Hewitt E, Yelifari L, Polderman AM, et al. (2000) Clinical epidemiology and classification of human oesophagostomiasis. Trans R Soc Trop Med Hyg 94: 177–182.
- Krepel HP, Baeta S, Polderman AM (1992) Human Oesophagostomum infection in northern Togo and Ghana: epidemiological aspects. Ann Trop Med Parasitol 86: 289–300.
- van Lieshout L, de Gruijter JM, Adu-Nsiah M, Haizel M, Verweij JJ, et al. (2005) Oesophagostomum bifurcum in non-human primates is not a potential reservoir for human infection in Ghana. Trop Med Int Health 10: 1315–1320.
- de Gruijter JM, Gasser RB, Polderman AM, Asigri V, Dijkshoorn L (2005) High resolution DNA fingerprinting by AFLP to study the genetic variation among *Oesophagostomum bifurcum* (Nematoda) from human and non-human primates from Ghana. Parasitol 130: 229–237.
- Krief S, Huffman M, Sévenet T, Guillot J, Bories C, et al. (2005) Non-invasive monitoring of the health of *Pan troglodytes schweinfurthii* in Kibale National Park, Uganda. Int J of Primatol 26: 467–490.
- Gillespie TR, Greiner EC, Chapman CA (2004) Gastrointestinal parasites of the guenons of western Uganda. J Parasitol 90: 1356–1360.
- Gillespie TR, Greiner EC, Chapman CA (2005) Gastrointestinal parasites of the colobus monkeys of Uganda. J Parasitol 91: 569–573.
- Anthony PP, McAdam I.W.J (1972) Helminthic pseudotumours of the bowel: thirty-fours cases of helminthoma. Gut 13: 8–16.
- Krief S, Vermeulen B, Lafosse S, Kasenene JM, Nieguitsila A, et al. (2010) Nodular worm infection in wild chimpanzees in western Uganda: a risk for human health? PLoS Negl Trop Dis 4: e630.
- Goldsmid J (1968) The differentiation of *Ternidens deminutus* and hookworm ova in human infections. Trans R Soc Tropl Med Hyg 62: 109–116.
- Gasser RB, Gruijter JMd, Polderman AM (2006) Insights into the epidemiology and genetic make-up of *Oesophagostomum bifurcum* from human and non-human primates using molecular tools. Parasitol 132: 453–460.
- Blotkamp J, Krepel H, Kumar V, Baeta S, Noordende J, et al. (1993) Observations on the morphology of adults and larval stages of *Oesophagostomum* sp. isolated from man in northern Togo and Ghana. J Helminthol 67: 49–61.
- Gasser RB, Woods WG, Huffman MA, Blotkamp J, Polderman AM (1999) Molecular separation of Oesophagostomum stephanostomum and Oesophagostomum bifurcum (Nematoda: Strongyloidea) from non-human primates. International J Parasitol 29: 1087–1091.
- Gasser RB, Chilton NB, Hoste H, Beveridge I (1993) Rapid sequencing of rDNA from single worms and eggs of parasitic helminths. Nucleic Acids Res 21: 2525–2526.
- Gasser RB, de Gruijter JM, Polderman AM (2009) The utility of molecular methods for elucidating primate-pathogen relationships - the Oesophagostonum bifurcum example. In: Huffman MA, Chapman CA, editors. Primate Parasite Ecology: The Dynamics and Study of Host-Parasite Relationships. Cambridge: Cambridge University Press.
- Chapman CA, Struhsaker TT, Lambert JE (2005) Thirty years of research in Kibale National Park, Uganda, reveals a complex picture for conservation. Int J of Primatol 26: 539–555.
- Watts DP (2012) Long-term research on chimpanzee behavioral ecology in Kibale National Park, Uganda. In: Kappeler PMaW, David P., editor. Long-Term Field Studies of Primates. Berlin, Heidelberg: Springer Berlin Heidelberg. pp. 313–338.
- 31. Goldberg TL, Paige SB, Chapman CA (2012) The Kibale EcoHealth Project: exploring connections among human health, animal health, and landscape dynamics in western Uganda In: Aguirre AA, Ostfeld RS, Daszak P, editors. New Directions in Conservation Medicine: Applied Cases of Ecological Health. New York: Oxford University Press. pp. 452–465.
- Garcia LS, Campbell J, Fritsche PTR, Hummert B, Johnston SP, et al. (2005) Procedures for the recovery and identification of parasites from the intestinal tract; Approved guideline—Second edition. Clinical and Laboratory Standard Institute. 109 p.
- Greiner EC, McIntosh A (2009) Collection methods and diagnostic procedures for primate parasitology. In: Chapman CA, Huffman MA, editors. Primate Parasite Ecology. Cambridge: Cambride Universit Press. pp. 3–29.
- Young KH, Bullock SL, Melvin DM, Spruill CL (1979) Ethyl acetate as a substitute for diethyl ether in the formalin-ether sedimentation technique. J Clin Microbiol 10: 852–853.
- W.H.O. (1991) Basic laboratory methods in medical parasitology. Geneva, Switzerland: World Health Organization.
- Romstad A, Gasser RB, Monti JR, Polderman AM, Nansen P, et al. (1997) Differentiation of Oesophagostomum bifurcum from Nectator americanus by PCR using genetic markers in spacer ribosomal DNA. Mol Cell Probes 11: 169–176.

- Yu S-K, Hu B, Deng Y, Li H-M, Ren W-X, et al. (2012) Phylogenetic studies of Oesophagostonum asperum from goats based on sequences of internal transcribed spacers of ribosomal deoxyribonucleic acid (DNA). Afr J Microbiol Res 6: 3360– 3365.
- Newton LA, Chilton NB, Monti JR, Bjorn H, Varady M, et al. (1997) Rapid PCR-based delineation of the porcine nodular worms, Oesophagostomum *dentatum* and *O. quadrispinulatum*. Mol Cell Probes 11: 149–153.
- Newton LA, Chilton NB, Beveridge I, Gasser RB (1998) Systematic relationships of some members of the genera *Oesophagostomum* and *Chabertia* (Nematoda: Chabertiidae) based on ribosomal DNA sequence data. Int J Parasitol 28: 1781– 1789.
- Campbell AJ, Gasser RB, Chilton NB (1995) Differences in a ribosomal DNA sequence of *Strongylus* species allows identification of single eggs. Int J Parasitol 25: 359–365.
- Chilton NB, Gasser RB (1999) Sequence differences in the internal transcribed spacers of DNA among four species of hookworm (Ancylostomatoidea: *Ancylostoma*). Int J Parasitol 29: 1971–1977.
- Larkin MA, Blackshields G, Brown NP, Chenna R, McGettigan PA, et al. (2007) Clustal W and Clustal X version 2.0. Bioinformatics 23: 2947–2948.
- Maddison W, Maddison DR (2001) Mesquite: a modular system for evolutionary analysis, version 2.75. Available: http://mesquiteproject.org/. Accessed 25 September 2012.
- Tamura K, Peterson D, Peterson N, Stecher G, Nei M, et al. (2011) MEGA5: molecular evolutionary genetics analysis using maximum likelihood, evolutionary distance, and maximum parsimony methods. Mol Biol Evol 28: 2731–2739.
- Hasegawa M, Kishino H, Yano T (1985) Dating of the human-ape splitting by a molecular clock of mitochondrial DNA. J Mol Evol 22: 160–174.
- Agresti A, Coull BA (1998) Approximate is better than "exact" for interval estimation of binomial proportions. Am Stat 52: 119–126.
- Rozsa L, Reiczigel J, Majoros G (2000) Quantifying parasites in samples of hosts. J Parasitol 86: 228–232.
- R Development Team (2008) R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing: Vienna, Austria, R Foundation for Statistical Computing.
- Paradis E, Claude J, Strimmer K (2004) APE: Analyses of Phylogenetics and Evolution in R language. Bioinformatics 20: 289–290.
- Orme CDL Freckleton RP, Thomas GH, Petzoldt T, Fritz SA, and Isaac NJB (2012), CAPER: comparative analyses of phylogenetics and evolution in R. Methods in Ecology and Evolution, 3: 145–151.
- Snaith TV, Chapman CA (2008) Red colobus monkeys display alternative behavioral responses to the costs of scramble competition. Behav Ecol 19: 1289– 1296.
- Janmaat KR, Olupot W, Chancellor RL, Arlet ME, Waser PM (2009) Longterm site fidelity and individual home range shifts in *Lophocebus albigena*. Int J Primatol 30: 443–466.
- Watts DP (2012) Long-term research on chimpanzee behavioral ecology in Kibale National Park, Uganda. Long-Term Field Studies of Primates: Springer. pp. 313–338.
- Pebsworth PA, MacIntosh AJ, Morgan HR, Huffman MA (2012) Factors influencing the ranging behavior of chacma baboons (*Papio hamadryas ursinus*) living in a human-modified habitat. Int J Primatol 33: 872–887.
- Barton RA, Whiten A, Strum SC, Byrne RW, Simpson AJ (1992) Habitat use and resource availability in baboons. Animal Behav 43: 831–844.
- Kaplin BA (2001) Ranging behavior of two species of guenons (*Cercopithecus lhoesti* and *C. mitis doggetti*) in the Nyungwe Forest Reserve, Rwanda. Int J Primatol 22: 521–548.
- Olupot W, Chapman CA, Brown CH, Waser PM (1994) Mangabey (*Cercocebus albigena*) population density, group size, and ranging: A twenty-year comparison. Am J Primatol 32: 197–205.
- Chapman CA, Chapman IJ (2000) Determinants of group size in primates: the importance of travel costs. In: Boinski S, Garber PA, editors. On the move: how and why animals travel in groups. Chicago: University of Chicago Press. pp. 24– 41.
- Oates JF (1977) The social life of a black-and-white colobus monkey, *Colobus guereza*. Zeitschrift fur Tierphysiologie 45: 1–60.
- Mitani JC, Sanders WJ, Lwanga JS, Windfelder TL (2001) Predatory behavior of crowned hawk-eagles (*Stephanoaetus coronatus*) in Kibale National Park, Uganda. Behav Ecol Sociobiol 49: 187–195.
- Ham RM (1994) Behaviour and ecology of grey-cheeked mangabeys (*Cercocebus albigena*) in the Lope Reserve, Gabon [Ph.D]: University of Stirling.
- Struhsaker TT (1981) Polyspecific associations among tropical rain-forest primates. Zeitschrift f
 ür Tierpsychologie 57: 268–304.
- Pontzer H, Wrangham RW (2004) Climbing and the daily energy cost of locomotion in wild chimpanzees: implications for hominoid locomotor evolution. J Human Evol 46: 315–333.
- Altmann J, Schoeller D, Altmann SA, Muruthi P, Sapolsky RM (1993) Body size and fatness of free-living baboons reflect food availability and activity levels. Am J Primatol 30: 149–161.
- Barreto ML, Genser B, Strina A, Teixeira MG, Assis AMO, et al. (2010) Impact of a citywide sanitation program in northeast Brazil on intestinal parasites infection in young children. Env Health Persp 118: 1637.
- 66. Slifko TR, Smith HV, Rose JB (2000) Emerging parasite zoonoses associated with water and food. Int J Parasitol 30: 1379–1393.

- Nunn CL, Altizer S, Jones KE, Sechrest W (2003) Comparative tests of parasite species richness in primates. Am Nat 162: 597–614.
- Davies AG (1994) Colobine populations. In: Davies AG, Oates JF, editors. Colobine monkeys: Their ecology, behaviour and evolution. Cambridge: Cambridge University Press. pp. 285–310.
- Clauss M (2003). Tannins in the nutrition of wild animals: a review. In Fidgett, AL, Clauss, M, Ganslosser U, Hatt, JM and Nijboer, J., eds. Zoo Animal Nutrition Vol. II. Filander: Fürth, pages 53–89.
- Freeland WJ (1976) Pathogens and the evolution of primate sociality. Biotropica 8: 12–24.
- Nunn CL, Altizer S (2006) Infectious diseases in primates: Behavior, ecology and evolution. Oxford: Oxford University Press.
- Ezenwa VO (2003) Habitat overlap and gastrointestinal parasitism in sympatric African bovids. Parasitol 126: 379–388.
- 73. Poulin R, Morand S (2004) Parasite biodiversity..Washington, D.C: Smithsonian Institution Press. 216 p.
- Kooriyama T, Hasegawa H, Shimozuru M, Tsubota T, Nishida T, et al. (2012) Parasitology of five primates in Mahale Mountains National Park, Tanzania. Primates 53: 365–375.
- Loehle C (1995) Social barriers to pathogen transmission in wild animal populations. Ecol 76: 326–335.
- Freeland WJ (1979) Primate social groups as biological islands. Ecol 60: 719– 728.

- McGrew WC, Tutin CEG, Collin DA, File SK (1989) Intestinal parasites of sympatric *Pan troglodytes* and *Papio* spp. at two sites Gombe (Tanzania) and Mt Assirik (Senegal). Am J Primatol 17: 147–155.
- Goldberg TL, Gillespie TR, Rwego IB, Estoff EE, Chapman CA (2008) Forest fragmentation as cause of bacterial transmission among primates, humans, and livestock, Uganda. Emerg Infect Dis 14: 1375–1382.
- Salyer SJ, Gillespie TR, Rwego IB, Chapman CA, Goldberg TL (2012) Epidemiology and molecular relationships of *Cryptosporidium* spp. in people, primates, and livestock from Western Uganda. PLoS Negl Trop Dis 6: e1597.
- Johnston AR, Gillespie TR, Rwego IB, McLachlan TLT, Kent AD, et al. (2010) Molecular epidemiology of cross-species *Giardia duodenalis* transmission in western Uganda. PLoS Negl Trop Dis 4: e683.
- Goldberg TL, Gillespie TR, Rwego IB, Wheeler E, Estoff EL, et al. (2007) Patterns of gastrointestinal bacterial exchange between chimpanzees and humans involved in research and tourism in western Uganda. Biol Conserv 135: 527–533.
- Utzinger J, Rinaldi L, Lohourignon LK, Rohner F, Zimmermann MB, et al. (2008) FLOTAC: a new sensitive technique for the diagnosis of hookworm infections in humans. Trans R Soc Trop Med Hyg 102: 84–90.
- Allen A, Ridley D (1970) Further observations on the formol-ether concentration technique for faecal parasites. J Clin Path 23: 545.